Author’s response to reviews

Title: The development and psychometric properties of a measure of clinician's attitudes to depression: the revised Depression Attitude Questionnaire (R-DAQ).

Authors:

Mark Haddad (mark.haddad.1@city.ac.uk)
Marco Menchetti (marco.menchetti3@unibo.it)
Eamonn McKeown (Eamonn.McKeown.1@city.ac.uk)
André Tylee (andre.tylee@kcl.ac.uk)
Anthony Mann (anthony.mann@kcl.ac.uk)

Version: 2
Date: 10 October 2014

Author's response to reviews:

Dear Rye,

Thank you for your helpful comments.

I have followed your directions and: added a title page with authors' details; added line numbering; provided detail of author contributions, and of competing interests (none), and of acknowledgements.

I have ensured that details about informed consent are provided, ensured that the figure (x 1) and tables (x 3) are referred to in the text. The tables are placed in correct order at the end of the manuscript, together with the figure title.